Author's response to reviews

Title: Recognizing Acute Delirium As part of your Routine [RADAR]: A validation study

Authors:

Philippe Voyer (philippe.voyer@fsi.ulaval.ca)
Nathalie Champoux (nathalie.champoux@umontreal.ca)
Johanne Desrosiers (johanne.desrosiers@usherbrooke.ca)
Philippe Landreville (philippe.landreville@psy.ulaval.ca)
Jane McCusker (jane.mccusker@mcgill.ca)
Johanne Monette (johanne.monette@mcgill.ca)
Maryse Savoie (maryse.savoie@vac-acc.gc.ca)
Sylvie Richard (sylvie.richard@fsi.ulaval.ca)
Pierre-Hugues Carmichael (pierre-hugues.carmichael.cha@ssss.gouv.qc.ca)

Version: 6 Date: 11 March 2015

Author's response to reviews:

Here were the editorial requirements:

• Clarify if additional files 1 and 2 are meant to be physically included in the main ms or just referenced at the end of the Introduction.
  o We think it is important to mention them in the introduction. But, they do not need to be physically included in the introduction.
  o It is important then that these two additional files are easily accessible at the end of the manuscript.

• Names of the ethics committees who approved the study need to be included.
  o They are now mentioned.

• Key words need to be removed from after the Acknowledgement.
  o They have been removed.